CrossMark

# 

**Citation:** Mutebi F, von Samson-Himmelstjerna G, Feldmeier H, Waiswa C, Bukeka Muhindo J, Krücken J (2016) Successful Treatment of Severe Tungiasis in Pigs Using a Topical Aerosol Containing Chlorfenvinphos, Dichlorphos and Gentian Violet. PLoS Negl Trop Dis 10(10): e0005056. doi:10.1371/journal.pntd.0005056

Editor: Aysegul Taylan Ozkan, Hitit University, Faculty of Medicine, TURKEY

Received: March 9, 2016

Accepted: September 20, 2016

Published: October 11, 2016

**Copyright:** © 2016 Mutebi et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Data Availability Statement:** All relevant data are within the paper and its Supporting Information files.

**Funding:** The present study was performed as a collaborative research project between Makarere University and the Institute of Parasitology and Tropical Veterinary Medicine, Freie Universität Berlin. The latter financed the study from internal funds and transferred money to Makerere University. FM received financial support from the German Academic Exchange Service (DAAD) while

RESEARCH ARTICLE

Successful Treatment of Severe Tungiasis in Pigs Using a Topical Aerosol Containing Chlorfenvinphos, Dichlorphos and Gentian Violet

Francis Mutebi<sup>1</sup>, Georg von Samson-Himmelstjerna<sup>2</sup>, Hermann Feldmeier<sup>3</sup>, Charles Waiswa<sup>1</sup>, Jeanne Bukeka Muhindo<sup>4</sup>, Jürgen Krücken<sup>2</sup>\*

1 School of Veterinary Medicine and Animal Resources, College of Veterinary Medicine, Animal Resources and Bio-security, Makerere University, Kampala, Uganda, 2 Institute for Parasitology and Tropical Veterinary Medicine, Freie Universität Berlin, Berlin, Germany, 3 Institute of Microbiology and Hygiene, Charité University Medicine, Berlin, Campus Benjamin Franklin, Berlin, Germany, 4 National Drug Authority, Kampala, Uganda

\* juergen.kruecken@fu-berlin.de

# Abstract

### Background

In endemic communities, zoonotic tungiasis, a severe skin disease caused by penetrating female sand fleas, is a public health hazard causing significant human and animal morbidity. No validated drugs are currently available for treatment of animal tungiasis. Due to the reservoir in domestic animals, integrated management of human and animal tungiasis is required to avert its negative effects.

### Methods and principal findings

A topical aerosol containing chlorfenvinphos 4.8%, dichlorphos 0.75% and gentian violet 0.145% licensed to treat tick infestations, myiasis and wound sepsis in animals in the study area, was tested for its potential tungicidal effects in a randomized controlled field trial against pig tungiasis in rural Uganda. Animals with at least one embedded flea were randomized in a treatment (n = 29) and a control (n = 26) group. One week after treatment, 58.6% of the treated pigs did not show any viable flea lesion whereas all control pigs had at least one viable lesion. After treatment the number of viable lesions (treated median = 0, overall range = 0–18 vs. control median = 11.5, range = 1–180) and the severity score for estimating acute pathology in pig tungiasis (treated median = 1, range = 0–3.5 vs. control median = 7, range = 0–25) were significantly lower in treated than in control pigs (p < 0.001). In the treatment group the median number of viable flea lesions decreased from 8.5 to 0 (p < 0.001). Similarly, the median acute severity score dropped from 6 to 1 (p < 0.001). Every pig in the treatment group showed a decrease in the number of viable fleas and tungiasis-associated acute morbidity while medians for both increased in the control group.

working on his PhD project. The funders, except of the co-authors GvSH and JK, had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing Interests:** The authors have declared that no competing interests exist.

### Conclusions

The study demonstrates that a topical treatment based on chlorfenvinphos, dichlorphos and gentian violet is highly effective against pig tungiasis. Due to its simplicity, the new approach can be used for the treatment of individual animals as well as in mass campaigns.

### Author Summary

Infection with the sand flea *Tunga penetrans* causes severe disease in humans and animals. There are no validated drugs for treatment of animal tungiasis preventing implementation of integrated tungiasis control interventions targeting human and animal infections. A field trial was conducted to evaluate the effects of a commercial insecticidal aerosol containing chlorfenvinphos 4.8%, dichlorphos 0.75%, and gentian violet 0.145%, which is licensed to treat tick infestations, myiasis and wound sepsis, on tungiasis in pigs, the major animal reservoir in rural hyperendemic villages in Uganda. Infected pigs were recruited and randomly assigned to treatment (n = 29) and control (n = 26) groups. Seven days after a single application of the aerosol onto the affected body parts, almost 60% of the treated pigs were cured while all control pigs had at least one penetrated sand flea. The number of viable sand fleas and the severity of the tungiasis were significantly lower in the treated pigs than in the controls. This demonstrates for the first time that a simple and effective topical treatment for animal tungiasis based on two organophosphate insecticides and an antibacterial agent can be used to cure individual animals and can be integrated in tungiasis control campaigns.

# Introduction

Tungiasis is a parasitosis of humans and animals caused by embedded female sand fleas belonging to the genus *Tunga*. So far 14 different *Tunga* species have been described [1-4] of which three have been reported in domestic animals: *Tunga hexalobulata*, *Tunga trimamillata* and *Tunga penetrans* [3, 5]. Only the latter two species are known to be zoonotic. Zoonotic tungiasis is a neglected tropical disease [6, 7] endemic in sub-Saharan Africa and Latin America. The disease is common among the poorest of the poor [3]. In endemic areas, tungiasis is characterized by seasonal fluctuations as well as a high prevalence and intensity in marginalized communities [8–15]. Sometimes, infections in humans and companion animals are also documented in non-endemic areas due to travelers coming from endemic areas [16–22].

With a size of about 1 mm in length, the sand flea is the smallest known member of the order Siphonaptera. After penetration, the female flea undergoes neosomy to attain a size of up to 13 mm in diameter in about two weeks. It remains viable for four to six weeks after which it dies *in situ* [23, 24]. While in the host skin, the burrowed flea provokes severe inflammation which in the majority of cases is complicated by secondary bacterial infections [25, 26]. The resultant morbidity is characterized by a wide range of manifestations and sequelae, some of which are potentially life-threatening [27–29].

Animal infections often occur alongside human infections but the main animal reservoir hosts vary between endemic areas [3, 30-33]. While dogs, cats and rodents have been reported as major animal hosts of sand fleas in South America [30, 31], pigs and, to some extent, dogs

were described as important animal reservoirs in Africa [32, 33]. However, there are only a few studies describing the role of animals in the epidemiology of tungiasis and it might well be that there are additional important species depending on socio-cultural and environmental conditions in the particular areas. Animal tungiasis is believed to amplify prevalence and infection intensity of human tungiasis and vice versa [31, 33]. The importance of an animal reservoir depends on the number of sand fleas it harbors since the number of eggs expelled is directly proportional to the infection intensity. Moreover, the degree of overlap between animal ranges and human activities is important since the mobility of fleas is not very high. Therefore, animals kept in or close to human houses can contaminate the immediate environment of the inhabitants with eggs that develop to larvae, pupae and finally host-seeking imagos.

In Bugiri District, Busoga, south eastern Uganda where the study was conducted, pigs were identified as the major animal reservoir for tungiasis and under poor husbandry they may carry several hundred embedded sand fleas during the peak of the transmission season [33]. This does not only cause severe disease in these animals [34] but also predisposes humans to infections through the shedding of eggs around or even in human dwellings [33].

Approaches aiming to achieve control of tungiasis in both, humans and domestic animals, are required for the efficient control of tungiasis in endemic communities [35]. In addition to proper animal husbandry practices, which are often difficult to implement in impoverished villages with very poor infrastructure, treatment protocols for infected animals would be highly beneficial. However, there is no approved drug treatment for animal tungiasis [36, 37]. Studies aimed at improving this situation are very scarce. A recent case report on goat tungiasis suggests that a topical spray containing chlorfenvinphos 4.8%, dichlorphos 0.75% and gentian violet 0.145% (Supona aerosol, previously Pfizer, now Zoetis, South Africa) has tungicidal properties [38]. The objective of this study was to systematically evaluate potential curative effects of the Supona aerosol on pig tungiasis. Supona aerosol is licensed to prevent wound infections and for treatment against ticks and blowfly maggots in cattle, sheep, goats, horses and dogs. The present study shows that a single treatment of pigs with Supona aerosol significantly reduced the intensity and severity of pig tungiasis within seven days.

# **Materials and Methods**

### Study area and population

The study was carried out in three parishes (Makoma, Wakawaka and Bulidha) of Bugiri District, Busoga, south eastern Uganda. Initially, pigs, which are the most important animal hosts of *T. penetrans* in the area [33], were examined. in ten villages These villages were purposively selected because they had a high prevalence of human and pig tungiasis [33]. All pigs in the villages were under a "back yard management system" whereby they were tethered under trees close to human houses but occasionally released to roam freely in search for food. Supplementation was occasionally done with household refuse and plant residues. Pigs which met the inclusion criteria were drawn from four villages, i.e. Busakira, Busindha and Masolya in Makoma Parish and Kibuye located in Wakawaka Parish [33]. The study villages are situated near each other and are characterized by the same physical and socio-economic characteristics. A detailed description of the study area was presented previously [33].

### Study design

The study had a non-blinded, parallel design with one treatment and one control group. For treatment, Supona aerosol containing chlorfenvinphos 4.8%, dichlorphos 0.75% and gentian violet 0.145% was sprayed on the affected body parts. The spray was applied once and the efficacy of the treatment was assessed after seven days. All examinations were carried out by the

same investigator (F.M.). The study was conducted between February 8, 2015 and March 17, 2015 which is the peak of the dry season when the attack rate of sand fleas is assumed to be high [39]. During the entire study period, the weather conditions remained unchanged (dry season with no rain). All pig-rearing households in ten villages were identified with the help of the Chairpersons of Local Council One, i.e. the political heads of the villages, or their representatives. After explaining the study objectives, an informed written consent was obtained from the pig owners and a census of pigs in the household was conducted. The feet of all pigs in the households were cleaned with water and a brush and examined systematically as described before [33]. Infected pigs were marked with ear tags and their biographic information was obtained by interviewing the pig owners since there were no records kept by the farmers. During pig examination, sand flea lesions (embedded sand fleas, mutilations of embedded fleas and scars remaining after death of the flea) were counted and staged according to the Fortaleza classification [23]. Clinical manifestations of tungiasis were described in detail for each topographic body area and recorded on individual forms. Photographs of pathological alterations were taken and their respective serial numbers were recorded for later review.

Clinical diagnosis of tungiasis was based on the features of the various stages as described before [23, 24]. Briefly, a dark spot of about 1–2 mm usually in the middle of a hyperemic area (stage II) or a round glassy to yellow often raised patch with a central dark spot of about 4–13 mm in diameter (stage III) characterize the viable stages. Viability was also assessed by recognized viability signs, namely expulsion of eggs, excretion of feces and release of a watery secretion on palpation of the lesions. Dead fleas present as circular and raised brown to black patches surrounded by necrotic tissue (stage IV) while an extruded sand flea is represented by an epidermal circular shallow crater with necrotic edges (stage V). Fresh sores with or without sand flea chitin remnants at the predilection sites were considered as mutilated sand flea lesions. The sand fleas collected from pigs in the study area were identified previously to be *T. penetrans* [33].

Only pigs with at least one viable sand flea (stage II-IIIa) and which were not expected to leave the study area for the next three weeks after the baseline examination were included in the study. Pigs meeting these inclusion criteria were randomly allocated in each household into two groups (control or treatment). Randomization at household level was chosen to exclude households as a confounding factor. A paper-based lottery system was used for the random allocation of treatments. If only one pig was present in the household, two papers reading 'treatment' and 'control' were placed in a bowel and one was drawn. If two or more pigs were present, ear tag numbers were drawn. If an uneven number of animals was included from the same household, an additional blank paper was included. In households with five pigs or more and a highly variable intensity of *T. penetrans* infection, pigs were stratified into two groups depending on the intensity of infection (i.e. light infection  $\leq 5$  lesions and heavy infection intensity > 5 lesions) before randomization. This procedure was chosen to exclude the possibility that heavily infected animals would by chance be allocated to the same group.

### Description of treatment details

The pigs in the treatment group were treated once topically with Supona aerosol by covering all the digits (with or without embedded sand fleas) up to the coronary band at the metacarpal and metatarsal joints of the respective limbs (S1 Fig). For lesions occurring at other topographic areas, the aerosol was only applied to sites with embedded sand fleas. No intervention was undertaken for the pigs in the control group until the end of the study period (day 7) after which they were also treated with Supona aerosol. Pig owners were requested to watch for any side effects among pigs after treatment, especially signs of acute toxicosis and anaphylaxis

which were explained to them, such as salivation, diarrhea, tremors, convulsions and collapse and to report them to the investigator as soon as detected. In case a pig could not be traced at the time of the visit, the household was revisited again on the same day or one day later.

### Evaluation of treatment effects

The primary outcome measure was the change in the number of viable embedded *T. penetrans*. In addition, the severity score for acute pig tungiasis (SSAPT) which was modified from a similar score for human tungiasis [40] and changes in this score were considered as a secondary outcome measure. Pigs were followed up only once, seven days after enrolment and treatment. The outcome measures were assessed at the individual animal level. For this purpose, the following parameters were determined: the number of viable sand fleas of the different stages, the number of morphological/topographic sites (in particular individual digits but for ectopic lesions also other body parts as detailed below for determination of the morbidity score) with viable sand fleas, the total number of sand fleas (viable plus already dead) and the SSAPT obtained as described below. The SSAPT allows the semi-quantitative assessment of morbidity using the clinical pathological features of tungiasis. Since acute morbidity correlates with the number of embedded sand fleas and decreases with reduction of the infection intensity, the SSAPT can be reliably used to quantify the severity of morbidity in a fairly objective manner [41–46]. In heavily infected pigs with mutilated lesions (S2 Fig) it was difficult to count the number of dead fleas and differentiating the mutilated lesions from those that had died naturally. In such cases only lesions which were distinct were analyzed.

The clinical variables used to quantify the SSAPT of the infected pigs were edema, hyperemia, ulcers, fissures, clusters of lesions, mutilation of lesions (as an indicator of itching), pain on digital pressure, ectopy of lesions, suppuration and/or abscessation and alteration of pig gait. The clinical features outlined above either involve a single or a cluster of lesions (e.g. local hyperemia) or affect the entire defined area (digit) or change the general behavior of the animal (e.g. limping). The digits were used as the primary units for the scoring of the morbidity because the majority of the lesions localized here. The pig distal limb was divided into four topographic sites constituted by each of the principal and accessory digits up to the distal metacarpal or metatarsal joints, hence a total of 16 digits for the four limbs of a pig. All other areas were considered ectopic sites of sand flea attachment. For signs that could be localized to each of the digits irrespective of extent, a score of one to three was assigned depending on the number of sites involved (Table 1). For other clinical manifestations, scores ranging from one to three were allocated depending on the relative significance of the sign on the wellbeing of the pig as proposed before [40]. For ectopic localizations, a score of 0.5 was allocated for every ectopic site up to a maximum of eight ectopic sites. Therefore, the maximum score (SSAPT) for an individual pig would be 27 (23+4). How the SSAPT is composed is depicted in Table 1.

### Ethics statement

The animal study adhered to the "Animals (Prevention of Cruelty) act", chapter 39, Constitution of Uganda and the National Drug Policy and Authority (Conduct of Ectoparasiticides Field Trials) Regulations, 2014. Ethical approval of the study was given by the ethical committee of the College of Veterinary Medicine, Animal Resources and Biosecurity, Makerere University (Ref. VAB/REC/14/101), the National Drug Authority (Ref. 462/NDA/DID/02/2016) and the National Council of Science and Technology (Ref. HS 1621). Written consent was obtained from pig owners. After the study all infected pigs in both groups which were still available were treated with Supona aerosol until all lesions were cleared.

#### Table 1. Severity score for acute pig tungiasis (SSAPT).

Characteristic clinical signs	Number of topographic sites affected	Score assigned
Hyperemia and/or edema <sup>a</sup>	1–5	1
	6–10	2
	11–16	3
Pain on digital pressure	1–5	1
	6–10	2
	11–16	3
Suppuration and/or abscess formation <sup>a</sup>	1–5	1
	6–10	2
	11–16	3
Clustering of lesions <sup>b</sup>	1–5	1
	6–10	2
	11–16	3
Fissure (s) <sup>a</sup>	1–5	1
	6–10	2
	11–16	3
Skin ulcerations <sup>a</sup>	1–5	1
	6–10	2
	11–16	3
Mutilation of lesions irrespective of sites involved <sup>c</sup>		2
Altered gait/lameness		3
Ectopy of lesions		0.5 <sup>d</sup>

<sup>a</sup>Irrespective of number of foci and size of the area involved on a designated topographical site

<sup>b</sup>Three or more lesions in close proximity (1–2 mm apart).

<sup>c</sup>Mutilation of lesion reflects intense itching

<sup>d</sup>For each ectopic discrete body part involved up to a maximum of eight ectopic sites; maximum 4 points

doi:10.1371/journal.pntd.0005056.t001

### Statistical analysis

Data was entered into Excel (Microsoft Office, 2007), double checked for errors which might have occurred during data entry and transferred to Stata Software package, Version 13 (Stata Corporation, College Station, Texas 77845 USA) and R 3.1.2 software for analysis. Descriptive statistics were generated. The PropCIs package in R was used to calculate 95% confidence intervals (CI) as Wilson score intervals. The binomial test was used to determine if frequencies of mutually exclusive outcomes were significantly different from 50% and the mid p exact test (as implemented in the epitools 0.5–7 package in R) was used to compare proportions between two groups. Since variables were not normally distributed, medians and range of values (minimum to maximum) are mainly provided. The Mann-Whitney U test was used to compare numerical data between groups while the Wilcoxon Matched Pairs Signed Rank Test was used to compare paired data from the same individual animal before and after treatment. Only pigs that were available at day 0 and day 7 after treatment were included in the statistical analyses. Spearman correlations coefficients were calculated with the cor.test command while logistic regression analysis was conducted using the glm function implemented in R software.

Generalized linear mixed models for numbers of viable embedded sand fleas (count data) and ordinal SSAPT data were obtained using a Bayesian approach as implemented in the MCMCglmm 2.22.1 R package. The MCMCglmm command was conducted to run a Monte-Carlo-Markov-Chain with 2 million iterations to estimate model parameters and a Poisson

model or an ordinal model were chosen for flea numbers and SSAPT, respectively. All other parameters in MCMCglmm were not changed from default values. Group (control vs. treatment) and time point (before vs. after treatment) were used as two levels of categorical variables and the animal number as random effect. Villages, sex and age of the pigs were considered as potential confounding independent variables. Household was not considered as an independent variable since randomization was performed at the household level and the number of households was relatively high (n = 39) compared to the number of pigs (n = 55). Variables were eliminated stepwise in order to improve the Deviance Information Criterion (DIC).

Effect sizes were calculated using the non-parametric Cliff's delta using a consistent estimation of the variance approach [47] with the function cliff.delta in the R package effsize 0.6.2. For this purpose, numbers of viable fleas or SSAPT values were compared (i) within the treatment group before and after treatment and (ii) between control and treatment group on day 7 post treatment.

Risk ratios describing the "risk to improve" (lower number of viable fleas or lower SSAPT) were calculated using the risk ratio function in the epitools R package with 95% CI determined by bootstrapping with 5000 iterations. A mid-p exact test was used to determine if the risk ratio was significantly different from 1.

#### Results

#### Description of the study groups

Out of 497 pigs examined in 141 households, 62 (12.5%) pigs met the inclusion criteria and were enrolled into the study. These pigs belonged to 23 households (16.3%). Overall, a median of 2 pigs and range of 1–6 pigs met the inclusion criteria per household. Of the enrolled pigs, 32 were randomly allocated to the treatment group and 30 to the control group. In the treatment group, three pigs were not available at follow up, and in the control group four pigs were lost. This limited the number of pigs included in the data analysis to 29 and 26, respectively. The number of pigs recruited per group did not differ significantly (p = 0.788 in a binomial test). Animal recruitment and randomization into groups is depicted in the flow diagram in S3 Fig. Table 2 summarizes the demographic characteristics of the two groups of pigs that were finally included in the analysis. Pigs in the treatment group were slightly older than those in the control group (p = 0.282 in a Man-Whitney U test). While the number of male pigs did not differ from that of the female pigs in the control group (p = 0.845), the treatment group had a higher number of females than males (p = 0.024 in a binomial test).

All pigs had lesions on the digits of the legs but six pigs also had lesions on ectopic sites such as skin along the metacarpal and metatarsals (n = 6), snout (n = 2), tail (n = 1) and testes (n = 1). Many lesions on the digits were mutilated and appeared either as fresh sores with chitin sand flea remnants or as necrotic hollow black ghosts. Viable lesions were mostly detected at the edges of the mutilated sites (S4 Fig). These were often accompanied by adjacent skin abrasions and/or ulcers at the predilection sites. Infected pigs had lesions at up to 21 distinct topographic sites.

### **Treatment effects**

No adverse effects were reported by the owners in any of the treated pigs and no pigs were excluded due to such effects. Before treatment, the number of viable fleas was high in both groups with a median of 8 (range 1–392) for the control group and a median of 9 (range 1–241) for the treatment group. The median SSAPT was 6.5 (range 3–24) in the control and 6 (range 4–17.5) in the treatment group. Following treatment, the number of viable and dead

#### Table 2. Demographic characteristics of the study groups of pigs included in the trial

Characteristic	Treatment group	Control group	p value
Total number of pigs recruited	29	26	0.788 <sup>c</sup>
Number of households <sup>a</sup>	20	18	0.871 <sup>c</sup>
Number of pigs included per household: median (range)	2 (1–3)	2 (1–3)	0.993 <sup>d</sup>
Village <sup>b</sup>			
Busindha	4	1	0.375 <sup>c</sup>
Masolya	9	7	0.804 <sup>c</sup>
Kibuye	11	12	1.000 <sup>c</sup>
Busakira	5	6 1.000 <sup>c</sup>	
Breed	All mixed breeds	All mixed breeds	
Sex			
Male	8	12 0.845 <sup>c</sup>	
Female	21	14 0.024 <sup>c</sup>	
Age in months: Median (range, minimum to maximum)	8 (2–24)	6 (2–18)	0.282 <sup>d</sup>

<sup>a</sup>To which the enrolled pigs belonged"

<sup>b</sup>To which the households with enrolled pigs belonged

<sup>c</sup>In a binomial test

<sup>d</sup>In a Mann-Whitney U test.

doi:10.1371/journal.pntd.0005056.t002

lesions dropped significantly (Table 3). The later was presumably attributed to exteriorization of the lesions (Fig 1 and S5 Fig). On day 7 after treatment, 58.6% (95% CI 40.74–74.49%) of the pigs in the treatment group did not show any viable lesion and were considered to be cured. The highest number of viable lesions recorded in the treatment group at day 7 was 18. This pig had 241 viable sand fleas before treatment. In contrast, all pigs in the control group had at least one viable lesion at day 7 (cure rate = 0%; 95% CI 0–12.87%) and one pig had up to 180 viable lesions (Table 3). Differences between the treatment group on day 7 and (i) the untreated control on day 7 and (ii) the treatment group at day 0 were highly significant (p < 0.001 in a

#### Table 3. Comparison of clinical characteristics of the two groups of pigs at baseline and after intervention.

	Control group (n = 26) Summary statistics		Treatment group (n = 29) Summary statistics		P-values			
Variable					control vs. treatment <sup>a</sup>		Day 0 vs. day 7 <sup>b</sup>	
	Day 0	Day 7	Day 0	Day 7	Day 0	Day 7	Control	Tratment
Number of animals with manipulated lesions (%)	25 (96.15%)	23 (88.5%)	29 (100%)	0 (0%)	0.473	p<0.001	0.360	<0.001
Median number of sites with viable lesions (range)	4.5 (1–20)	6 (1–20)	6 (1–21)	0 (0–7)	0.445	p<0.001	0.447	<0.001
Median number of lesions per animal (range)	14 (1–832)	21.5 (1–838)	18 (1–756)	9 (0–120)	0.367	0.046	0.038	<0.001
Median number of viable lesions (range)	8 (1–392)	11.5 (1–180)	9 (1–241)	0 (0–18)	0.473	p<0.001	0.807	<0.001
Stage II: median (range)	0 (0–4)	0 (1–11)	0 (0–19)	0 (0–4)	0.463	0.125	0.832	0.273
Stage IIIa: median (range)	8 (1–392)	9.5 (1–177)	8 (1–241)	0 (0–14)	0.543	p<0.001	0.931	<0.001
Stage IIIb: median (range)	0 (0–17)	0 (0–11)	0 (0–8)	0 (0 (0–6)	0.826	0.078	0.63	0.14
Median number of dead lesions per animal (range)	6.5 (0–440)	9.5 (0–702)	10 (0–515)	6 (0–102)	0.348	0.715	0.006	0.516
Median acute morbidity score (range)	6 (3–24)	7 (0–25)	6 (1–17.5)	1 (0–3.5)	0.574	p<0.001	0.536	<0.001

<sup>a</sup>Calculated using the mid-p exact test for percentages and the unpaired Mann-Whitney U test for comparison of medians between groups at the same timepoint.

<sup>b</sup>Calculated using the mid-p exact test for percentages and the Wilcoxon Matched Pairs Signed Rank Test for comparison of medians between groups at the same time-point.

doi:10.1371/journal.pntd.0005056.t003



**Fig 1. Left hind leg digits of a pig before (A) and after (B) treatment.** This pig had 297 lesions before treatment (208 viable and 89 dead) of which 6 were ectopic on metacarpal and metatarsal skin. Only three viable lesions were detectable after treatment. Selected viable lesions are marked with black arrows while some dead lesions are marked with white arrows.

doi:10.1371/journal.pntd.0005056.g001

Mann-Whitney-U test and a Wilcoxon Matched Paired Signed Rank test). Because of massive mutilation of lesions among the untreated pigs by scratching the affected sites (Fig 2 and S6 Fig), the number of viable lesions slightly decreased in some pigs but overall, there was an increment in the number of viable lesions and in the SSAPT in the control group (Fig 3). In contrast, there was a highly significant reduction in the number of viable penetrated fleas and in the SSAPT for the treatment group. In fact, both values decreased for every individual pig after treatment (Fig 3).

Using logistic regression, no significant effect of the number of viable fleas at baseline on complete clearance was detectable (p = 0.224). However, the median number of viable fleas at baseline was 7.5 (range 1–241) in those animals that had completely cleared the infection on day 7 after treatment but was 34.5 (range 4–208) in the animals that still had at least one viable flea. This difference was statistically significant (p = 0.012 in a Mann-Whitney U test). The animal with the highest number of viable fleas before treatment (241) was completely cured by treatment.



Fig 2. Digits of a pig in the control group at baseline (A) and after seven days (B). This pig had 29 lesions of which 21 were viable (black arrows) and eight were dead at baseline but one week later there were 17 viable lesions and more than 28 dead lesions many of which were mutilated (white arrows).

doi:10.1371/journal.pntd.0005056.g002





doi:10.1371/journal.pntd.0005056.g003

Coefficient	Posterior mean	95% Confidence interval	P value (MCMC)
Intercept	2.293	1.726–2.845	<0.001
Day 7 <sup>a</sup>	0.106	-0.262–0.478	0.570
Treatment group <sup>a</sup>	0.142	-0.628–0.924	0.718
Day 7*Treatment group	-2.697	-3.2322.082	<0.001
DIC	619.6		

Table 4. Generalized linear mixed regression model for changes in number of viable fleas calculated using a Monte-Carlo-Markov-Chain (MCMC) approach.

<sup>a</sup>Reference categories are day 0 and the control group.

doi:10.1371/journal.pntd.0005056.t004

Overall, there was a high correlation between the number of viable flea lesions and the SSAPT (Spearman's rho = 0.731, p< 0.001) indicating that severity of disease increased with numbers of embedded fleas and that both parameters are well suited to determine potential effects of treatment.

#### **Regression analyses**

In order to determine the effects of potential confounding variables, the paired datasets (two groups  $\times$  two time points) were analyzed using generalized linear mixed models. The considered independent variables age, sex and village consistently dropped out of all models since they always increased the DIC.

For numbers of viable fleas (count data), the best model with the lowest DIC included the variables group and time point and their interaction (Table 4). Only the interaction between treatment and time point on day 7 after treatment had a significant effect on numbers of embedded sand fleas. The average reduction in the number of viable fleas was approximately 15 ( $e^{2.697}$ ) in the treatment group on day 7 compared to the reference categories control group and day 0. This difference was highly significant (p<0.001).

With regard to the ordinal SSAPT data, a model containing only the variables group and time point was again clearly preferred according to the DIC. This model also showed that significant effects were only observed in the combination of treatment with the time point day 7 after treatment resulting in an average decrease in the SSAPT of 1.44 in this combination (Table 5).

#### Effect sizes

Effect sizes for treatment with Supona aerosol were calculated non-parametrically as Cliff's delta values. For the number of viable fleas, a comparison in the treatment group before and after treatment provided a Cliff's delta of 0.76 (95% CI: 0.53–0.89) and comparison of control

Table 5.	Generalized linear mixed regression model for changes in the severity score for acute pig tungiasis (SSAPT) calculated using a Monte-
Carlo-Ma	arkov-Chain (MCMC) approach.

Coefficient	Posterior mean	95% Confidence interval	P value (MCMC)
Intercept	0.884	0.616–1.438	<0.001
Day 7 <sup>a</sup>	0.176	-0.396–0.498	0.524
Treatment group <sup>a</sup>	-0.056	-0.537–0.260	0.871
Day 7*Treatment group	-1.441	-2.1000.538	<0.001
DIC	636.2		

<sup>a</sup>Reference categories are day 0 and the control group.

doi:10.1371/journal.pntd.0005056.t005

and treatment group on day 7 had a similar value of 0.76 (95% CI: 0.55–0.88). For the SSAPT, the comparison of values for the treatment group before and after treatment led to a Cliff's delta of 0.97 (95% CI: 0.93–0.98) while the comparison between control and treatment group was somewhat lower with a Cliff's delta of 0.79 (95% CI: 0.54–0.91). Since all Cliff's delta values  $\geq$ 0.474 are generally considered as "large" effects [48], all calculations indicate that the effects of a single Supona aerosol treatment on the number of viable fleas or the morbidity score are not only significant but also large.

In order to determine how different the probabilities of pigs in the two groups were to improve (decreased number of viable fleas or decreased SSAPT) between both visits, risk ratios were calculated. In the treatment group, the risk ratio to have a decreased number of viable fleas compared to the control group was 2.36 (95% CI: 1.63–4.33; p<0.001) in a mid-p exact test). Similarly, the risk ratio to show a reduced SSAPT in the comparison between treatment and control group was 2.89 (95% CI: 1.86–6.50, p<0.001).

## Discussion

The present study shows for the first time curative efficacy of a drug against pig tungiasis in a randomized, controlled trial using previously validated outcome measures, namely viability of embedded sand fleas and an acute morbidity score. Such an approach has never been used in animal tungiasis. The results clearly show that treatment with the Supona aerosol significantly reduced the number of viable embedded sand fleas and tungiasis-associated morbidity. The large effects sizes as determined by Cliff's delta for both, number of viable fleas and SSAPT, clearly show that the treatment had not only a statistically significant effect but considerably improved the health of the pigs. The risk ratios above 2 furthermore show that improvement was observed much more often in the treatment group than in the control group. In fact, every single treated animal had a decrease in the number of viable fleas and in the SSAPT.

The duration of the study was limited by the end of the dry season but a second visit after 14 days would have been possible for many animals included in the study. However, this was not done since there was a considerable problem with compliance of farmers owning pigs in the control group. Seeing that the health status of the animals in the treatment group had improved so much, they claimed that the animals in the control group should also be treated. For this reason and due to considerations regarding animal welfare, all animals were treated at the second visit.

To date, only a few drugs have been tested to treat animal tungiasis with variable outcomes. The most important classes of ectoparasiticides on the market, macrocyclic lactones and pyrethroids, would be considered obvious candidates to treat animal tungiasis. In particular, the endectocide ivermectin, which would also cure several other frequently co-endemic parasitoses such as scabies, ascariosis, human hookworm disease as well as pig infections with strongylid gastrointestinal nematodes, is an obvious candidate to be evaluated for its effects against *T. penetrans.* Indeed, ivermectin has been used several times to treat tungiasis.

A single case report of dog tungiasis from Denmark suggested that ivermectin is an effective drug [49]. However, the dog was treated subcutaneously with ivermectin, which is not licensed for treatment of dogs due to severe adverse effects in mdr-1 deficient animals. It is most important to note that the authors reported only that lesions resolved within one month, a time span when most lesions would have resolved without treatment anyhow. A randomized controlled trial found only minimal effects of oral ivermectin treatment on embedded sand fleas in humans [50].

In a case serious on pigs in Tanzania, Cooper applied lindane (hexachlorocyclohexane), an organochloride on *T. penetrans* lesions. The author claimed that lindane was effective but no

data was presented to support the statement [51, 52]. Topical applications of an oily solution of Trichlorphone 0.2% (Neguvon) applied on embedded sand fleas in dog foot pads was also reported to be tungicidal [53]. Again, no convincing data was provided. In another attempt to treat tungiasis in pets, neither Neguvon nor a collar containing a combination of propoxur and flumethrin (Kiltix) were successful [35]. A variety of other drugs such as niridazole [54], albendazole, thiabendazole [55], trichlorphone and potassium permanganate [37, 43] have been used in humans to treat tungiasis but all these were proven to have negligible effects on embedded sand fleas.

Another field trial reported a reduction in the number of sand fleas in dogs following application of a spot on in the neck with a formulation containing 10% imidacloprid and 50% permethrin (Advantix) [56]. The study was not conceived as a randomized control trial. The reduction of the number of embedded sand fleas was very slow and no data were provided on the viability of the embedded sand fleas. It seems that only new infections were prevented, presumably due to repellent and not due to insecticidal effects of the product and the already embedded sand fleas persisted and died *in situ* at the end of their normal life span. Alternatively, the slow onset of the drug effect might be caused by a slow distribution from the neck, where the drug was applied, to the footpad, where sand fleas are embedded. Moreover, the number of lesions re-increased already three weeks after treatment indicating that the drug did not protect against newly invading *T. penetrans* for four weeks as it does against other flea species such as the cat flea *Ctenocephalides felis* [57].

In the current field trial, the drug was applied directly on the lesions. This obviously achieved a local concentration of the active ingredient lethal to the embedded sand fleas. The dual action of the formulation-killing the embedded sand fleas and acting as an antimicrobial agent-presumably contributed to the quick reduction of inflammation, thus facilitating the shedding of dead fleas and repair of tissue damage. This explains the rapid reduction in the severity score for acute tungiasis of the treated pigs. The persistence of a few viable lesions following treatment may have resulted from failure of the chemical reaching the embedded sand fleas, e.g. due to the necrotic debris on the skin. Indeed, the median number of embedded viable fleas before treatment was significantly higher in those pigs that still had at least one viable flea after treatment. However, the number of viable fleas was not a significant variable in the logistic regression analysis suggesting that additional parameters influence the effects of the drugs.

It has been demonstrated before in humans that the intensity of infection correlates strongly with the clinical severity of tungiasis [39]. Measures of acute morbidity (SSAPT) dramatically decreased with a reduced intensity of infection, whereas measures of chronic morbidity only slightly change over time, since the latter reflect the severity of previous tungiasis episodes [45]. Therefore, changes in acute morbidity scores are an appropriate outcome measure to determine the efficacy of interventions against tungiasis. Since Supona aerosol treatment strongly decreased the number of viable lesions and concomitantly ameliorated acute morbidity, conclusions of the study are strongly supported by the data despite the lack of precision regarding the number of lesions in some highly infected animals.

The results of the present study suffer from three major limitations: (i) the difficulties to count the number of all lesions and viable lesions in clusters, (ii) the loss of initially viable lesions due to mutilation by the pigs and (iii) the short duration of the study. Clustering of lesions is a common feature of animal and human tungiais [25, 34]. Pruritus of lesions is often associated with extensive mutilation of lesions, and abrasions or ulcerations of the skin, resulting from intense rubbing of affected sites against objects. Mutilations of lesions and clustering of lesions are inevitable problems in a randomized controlled trial which limit the evaluation of individual lesions. Severity of tungiasis appears to increase with the size of the neosome, the

infection intensity and the associated clustering of lesions [25]. In severe cases, it is therefore likely that the number of lesions is underestimated which can also lead to underestimation of the effects of the drug. Although only semi-quantitative, the acute morbidity score is a useful alternative to the number of viable lesions to determine the effects of drug treatment. While all the treated pigs did not show any evidence of itching or pain as indicated by absence of new skin abrasions or lesion excoriations, almost all untreated pigs had persistent or newly mutilated lesions, abrasions or skin ulcerations at the end of the study.

Regarding the third limitation, the short duration of the follow-up period, a longer follow up period as well as inclusion of a third group with multiple treatments compared to a groups with a single treatment would have been desirable. However, this study was a follow up to a larger study [33] and there were no extramural funds available to perform additional field work. Moreover, as mentioned above, pig owners were usually not willing to leave their pigs untreatment throughout the study period. Thus, analysis of long-term effects and determination of an optimized treatment protocol with a recommended application frequency to maximize protective effects while minimizing the costs for the farmers is a subject of future studies.

In Uganda, Supona aerosol is commonly used as a wound antiseptic in animals, and it is applied to prevent wound sepsis, myiasis and tick infestations. Chlorfenvinphos and dichlorphos are organophosphates that are used as acaricides and insecticides with cholinesterase inhibitory effects while gentian violet is a triarylmethane compound with antifungal, antibacterial as well as anthelminthic properties. A combination of antibacterial and insecticidal properties in addition to the ease of application appear to make the Supona aerosol a good choice for treatment of tungiasis in animals since secondary infections with bacteria are common in tungiasis. Effects of control measures in domestic animals can be expected to reduce the burden of human tungiasis as well hence animal treatment should be considered at least in outbreak situations when money for control is available from public health and governmental institutions. Quantitative studies comparing human and animal tungiasis between intervention and control villages are required to demonstrate such effects. Chemical control of animal tungiasis should be integrated with other control practices such as public health education, animal confinement, hygienic housing and treatment of humans.

A complete blinding during the study was not possible since it would have required different people doing assessment/treatment and evaluation of effects but there were not enough personnel to do that. Initially, it was intended to conduct the study with blinding of the examiner in a way that he would not have access to the records from the intervention day during the follow-up visit. However, the differences in flea numbers and in particular in morbidity were so obvious that is was immediately clear to which group the animal belonged. Of course, the missing blinding might have introduced bias into the study. However, the quantitative differences between both groups were so large that the general conclusions can be considered to be valid.

#### Conclusions

The study has demonstrated a therapy using a simple, effective and rapidly acting topical drug against severe tungiasis in livestock. The combined antibacterial effects of gentian violet and the ectoparasitic effects of chlorfenvinphos plus dichlorphos make Supona aerosol a prototype formulation for treatment of animal tungiasis. The formulation can be used to treat individual cases and has the potential to be used in systematic treatment of infected animals to eliminate *T. penetrans* animal reservoirs in the endemic communities with minimal technical and financial resource inputs. As such these findings are of major importance for future One Health based tungiasis control initiatives which are urgently needed in the endemic regions.

## **Supporting Information**

**S1 Fig. Pig digits with several sand flea lesions covered with Supona aerosol.** All digits of pigs in the treatment group, irrespectively of whether they had sand fleas or not, were covered with the spray (purple coloration). (PDF)

S2 Fig. Pig digits with several sand fleas some of which are mutilated by the animal (black arrows).

(PDF)

S3 Fig. Flow diagram describing animal recruitment, allocation into groups and losses during the study according to the guidelines of the CONSORT statement (<u>http://www.consort-statement.org/</u>).

(PDF)

S4 Fig. Heavy infection and extensive mutilation of sand flea lesions on all digits with viable (arrows) sand fleas evident at the edges of the mutilated and necrotic areas. (PDF)

**S5 Fig. Digits of the left front leg before (A) and after treatment (B).** This pig had 180 viable (black arrows) and another 396 dead lesions (white arrows) before treatment. On day 7 after treatment, only seven viable embedded sand fleas were detected. (PDF)

**S6 Fig. Right front leg of a pig in the control group at baseline (A) and 7 days later (B).** At baseline, this pig had 49 viable and 35 dead lesions but a week later it had a total of 181 discernible lesions of which 98 were viable (black arrows) while 83 were mutilated (white arrows). (PDF)

### Acknowledgments

Field activities were conducted with support from Mr. Aponera Oloka and Chairpersons of the respective Local Council One. We are grateful to National Drug Authority and the District Veterinary Office of Bugiri for their support. High Heights Initiatives coordinated the field activities. Data is part of a PhD thesis by Francis Mutebi.

### **Author Contributions**

Conceived and designed the experiments: FM GvSH HF JK JBM CW.

Performed the experiments: FM.

Analyzed the data: FM JK.

Wrote the paper: FM JK GvSH HF CW JBM.

#### References

- 1. Linardi PM, Avelar DMD. Neosomes of tungid fleas on wild and domestic animals. Parasitiol Res. 2014; 113:3517–33. doi: 10.1007/s00436-014-4081-8 PMID: 25141814
- Linardi PM, Beaucourn J-C, Avelar DMd, Belaz S. Notes on the genus Tunga (Siphonaptera: Tungidae) II–neosomes, morphology, classification, and other taxonomic notes. Parasite. 2014; 21(68):1– 17. doi: 10.1015/parasite 2014067 PMID: 25514594

- Pampiglione S, Fioravanti ML, Gustinelli A, Onore G, Montovani B, Luchetti A, et al. Sand flea (*Tunga* spp.) infections in man and domestic animals: state of the art. Med Vet Entomol. 2009; 23(3):172–86. doi: 10.1111/j.1365-2915.2009.00807.x PMID: 19712148
- Ezquiaga MC, Linardi MP, De Avelar DM, Lareschi M. A new species of *Tunga* perforating the osteoderms of its armadillo host in Argentina and redescription of the male of *Tunga penetrans*. Med Vet Entomol. 2015; 29(2):196–204. doi: 10.1111/mve.12106 PMID: 25630228
- De Avelar DM, Filho EJF, Linardi PM. A new species of tunga (siphonaptera: Tungidae) parasitizing cattle from Brazil. J Med Entomol. 2013; 50(4):679–84. doi: 10.1603/me12221 PMID: 23926764
- Heukelbach J, Oliveira FAS, Hesse G, Feldmeier H. Tungiasis: A neglected health problem of poor communities. Trop Med Int Health. 2001; 6:267–72. doi: 10.1046/j.1365-3156.2001.00716.x PMID: 11348517
- Karunamoorthi K. Tungiasis: A neglected epidermal parasitic skin disease of marginalized populations —A call for global science and policy. Parasitol Res. 2013; 112(10):3635–43. doi: 10.1007/s00436-013-3551-8 PMID: 23949241
- Ade-Serrano MA, Ejezie GC. Prevalence of tungiasis in Oto-ijanikin village, Badagry, Lagos State, Nigeria. Ann. Trop.Med. Parasitol. 1981; 75:471–2. doi: 10.1080/00034983.1981.11687469 PMID: 7305515
- Carvalho RWD, Almeida ABD, Barbosa-Silva SC, Amorim M, Ribeiro PC, Serra-Freire NM. The patterns of tungiasis in Araruama Township, State of Rio de Janeiro, Brazil. Mem Inst Oswaldo Cruz. 2003; 98(000–000):1–6.
- Chadee DD. Tungiasis among five communities in South Western Trinidad, West Indies. Ann. Trop. Med. Parasitol. 1998; 92:107–13. doi: 10.1080/00034989860238 PMID: 9614460
- Chadee DD, Furlonge E, Naraynsingh C, Maitre LA. Distribution and prevalence of Tunga penetrans in Coastal South Trinidad, West Indies. Trans R Soc Trop Med Hyg. 1991; 85:549. doi: 10.1016/0035-9203(91)90253-u PMID: 1755071
- Mazigo HD, Bahemana E, Konje ET, Dyegura O, Mnyone LL, Kweka EJ, et al. Jigger flea infestation (tungiasis) in rural western Tanzania: High prevalence and severe morbidity. Trans R Soc Trop Med Hyg. 2012; 106(4):259–63. doi: 10.1016/j.trstmh.2011.12.001 PMID: 22305586
- Muehlen M, Heukelbach J, Wilcke T, Winter B, Mehlhorn H, Feldmeier H. Investigations on the biology, epidemiology, pathology and control of *Tunga penetrans* in Brazil II. Prevalence, parasite load, topographic distribution of lesions in the population of a traditional fishing Village. Parasitol Res. 2003; 90:449–55. doi: 10.1007/s00436-003-0877-7 PMID: 12768415
- 14. Ugbomoiko US, Ofoezie I, Heukelbach J. Tungiasis: high prevalence, parasite load, and morbidity in a rural community in Lagos State, Nigeria. Int J Dermatol. 2007; 46:475–81. doi: 10.1111/j.1365-4632. 2007.03245.x PMID: 17472674
- Wilcke T, Heukelbach J, Saboia-Moura RC, Kerr-Pontes RL, Feldmeier H. High Prevalence of tungiasis in a poor neighbourhood in Fortaleza, North Eastern Brazil. Acta Trop. 2002; 83:255–8. doi: 10. 1016/s0001-706x(02)00133-x PMID: 12204399
- Akwee J, Ivonye C, Henriques-Forsythe M, Foungeh T. Imported jigger. Am J Case Rep. 2008; 9:382–4.
- Arranz J, Taberner R, Llambrich A, Nadal C, Torne I, Vila A, et al. Four imported cases of tungiasis in Mallorca. Travel Med Infect Dis. 2011; 9(3):161–4. doi: 10.1016/j.tmaid.2011.03.005 PMID: 21530410
- 18. Chieh-Wen C, Haw-Yueh T, Shiou-Hwa J. Tungiasis: A case report and review of literature. Dermatol Sin. 2011; 29:29–31. doi: 10.1016/j.dsi.2011.01.004
- Sanusi ID, Brown EB, Shepard TG, Grafton WD. Tungiasis: Report of one case and review of the 14 reported cases in the United States. J Am Acad Dermatol. 1989; 20:941–4. doi: 10.1016/s0190-9622 (89)70115-8 PMID: 2654224
- Spradbery JP, Bromley J, Dixon R, Tetlow L. Tungiasis in Australia: an exotic disease threat. Med J Aust. 1994; 161(2):173. PMID: 8028549.
- Veraldi S, Carrera C, Schianchi R. Tungiasis has reached Europe. Dermatology. 2000; 201:382. doi: 10.1159/000051568 PMID: 11146364
- Wardhaugh AD, Norris JF. A case of imported tungiasis in Scotland initially mimicking Verrucae vulgaris. Scott Med J. 1994;(39: ):146–7. PMID: 8778973
- Eisele M, Heukelbach J, Marck EV, Mehlhorn H, Meckes O, Franck S, et al. Investigations on the biology, epidemiology, pathology and control of *Tunga penetrans* in Brazil: I. Natural history of tungiasis in man. Parasitol Res. 2003; 90:87–99. doi: 10.1007/s00436-002-0817-y PMID: 12756541
- 24. Feldmeier H, Witt L, Schwalfenberg S, Linardi PM, Ribeiro RA, Capaz RAC, et al. Investigations on the biology, epidemiology, pathology and control of Tunga penetrans in Brazil. VI. Natural history of the

infestation in laboratory-raised Wistar rats. Parasitol Res. 2007; 102:1–13. doi: 10.1007/s00436-007-0731-4 PMID: 17851690

- Feldmeier H, Eisele M, Marck EV, Mehlhorn H, Ribeiro R, Heukelbach J. Investigations on the biology, epidemiology, pathology and control of *Tunga penetrans* in Brazil: IV. Clinical and Histopathology. Parasitol Res. 2004; 94:275–82. doi: 10.1007/s00436-004-1197-2 PMID: 15368123
- Feldmeier H, Heukelbach J, Eisele M, Sousa AQ, Barbosa LM, Carvalho CB. Bacterial superinfections in human tungiasis. Trop Med Int Health. 2002; 7:559–64. doi: 10.1046/j.1365-3156.2002.00904.x PMID: 12100437
- Feldmeier H, Eisele M, Saboia-Moura RC, Heukelbach J. Severe tungiasis in underprevileged communities: Case series from Brazil. Emerg Infect Dis. 2003; 9(8):949–55. doi: 10.3201/eid0908.030041 PMID: 12967492
- 28. Tonge BL. Tetanus from chigger flea sores. J Trop Pediatr. 1989; 35:94. PMID: 2724405
- 29. Jawoko K. Jiggers outbreak in Uganda. CMAJ. 2011; 183(1):E33–E4. doi: 10.1503/cmaj.109-3726 PMID: 21115678
- Heukelbach J, Costa AM, Wilcke T, Mencke N, Feldmeier H. The animal reservoirs of *Tunga penetrans* in severely affected communities of North East Brazil. Med Vet Entomol. 2004; 18:307–13.
- Pilger D, Schwalfenberg S, Heukelbach J, Witt L, Mehlhorn H, Mencke N, et al. Investigations on the biology, epidemiology, pathology, and control of Tunga penetrans in Brazil: VII. The importance of animal reservoirs for human infestation. Parasitol Res. 2008; 102(5):875–80. doi: <u>10.1007/s00436-007-</u> 0840-0 PMID: 18172688
- Ugbomoiko US, Ariza L, Heukelbach J. Pigs are the most important animal reservoir for Tunga penetrans (jigger flea) in rural Nigeria. Trop Doct. 2008; 38(4):226–7. Epub 2008/09/30. doi: 10.1258/td. 2007.070352 PMID: 18820191.
- Mutebi F, Krücken J, Feldmeier H, Waiswa C, Mencke N, Sentongo E, et al. Animal reservoirs of zoonotic tungiasis in endemic rural villages of Uganda. PLoS Negl Trop Dis. 2015; 9(10):e0004126. doi: 10.1371/journal.pntd.0004126 PMID: 26473360
- Mutebi F, Krücken J, Feldmeier H, Waiswa C, Mencke N, Samson-Himmelstjerna Gv. Tungiasis-associated morbidity in pigs and dogs in endemic villages of Uganda. Parasit Vectors. 2016; 9(44):1–9. doi: 10.1186/s13071-016-1320-0 PMID: 26817587
- **35.** Pilger D, Schwalfenberg S, Heukelbach J, Witt L, Mencke N, Khakban A, et al. Controlling tungiasis in an impoverished Community: An Intervention Study. PLoS Negl Trop Dis. 2010; 2(10):e324.
- Marin RE, Houston R, Omanska-Klusek A, Alcaraz A, Garcia JP, Uzal FA. Pathology and diagnosis of proliferative and ulcerative dermatitis associated with *Tunga penetrans* infestation in cattle. J Vet Diagn Invest. 2015; 27(1):80–5. doi: 10.1177/1040638714559597 PMID: 25525141
- Heukelbach J. Revision on tungiasis: Treatment options and prevention. Expert Rev Anti-Infe. 2006; 4 (1):151–7. doi: 10.1586/14787210.4.1.151 PMID: 16441216
- Mutebi F, Krücken J, Mencke N, Feldmeier H, von Samson-Himmelstjerna G, Waiswa C. Two Severe Cases of Tungiasis in Goat Kids in Uganda. J Insect Sci. 2016; 16(1): pii 34 doi: <u>10.1093/jisesa/iew016</u> PMID: 27012871
- Heukelbach J, Wilcke T, Harm G, Feldmeier H. Seasonal variation of tungiasis in an endemic community. Am J Trop Med Hyg. 2005; 72:145–9. PMID: 15741550
- Kehr JD, Heukelbach J, Mehlhorn H, Feldmeier H. Morbidity assessment in sand flea disease (tungiasis). Parasitol Res. 2007; 100:413–21. doi: 10.1007/s00436-006-0348-Z PMID: 17058108
- Buckendahl J, Heukelbach J, Ariza L, Kehr JD, Seidenschwang M, Feldmeier H. Control of tungiasis through intermittent application of a plant-based repellant: An intervention study in a resource-poor community in Brazil. PLoS Negl Trop Dis. 2010; 4(11/e879):1–9. doi: 10.1371/journal.pntd.0000879 PMID: 21085467
- Feldmeier H, Kehr JD, Heukelbach J. A plant based repellant protects against *Tunga penetrans* infestation and sand flea disease. Acta Trop. 2006; 99:126–36. doi: 10.1016/j.actatropica.2006.05.013 PMID: 17010927
- Schwalfenberg S, Witt LH, Kehr JD, Feldmeier H, Heukelbach J. Prevention of tungiasis using a biological repellant: a small case series. Ann Trop Med Parasitol. 2004; 98(1):89–94. doi: <u>10.1179/</u>000349804225003091 PMID: 15000736
- Thielecke M, Nordin P, Ngomi N, Feldmeier H. Treatment of tungiasis with dimeticone: A proof-of principle study in rural Kenya. PLoS Negl Trop Dis. 2014; 8(7):e3058. doi: 10.1371/journal.pntd.0003058 PMID: 25079375
- Thielecke M, Raharimanga V, Rogier C, Stauss-Grabo M, Richard V, Feldmeier H. Prevention of Tungiasis and Tungiasis-Associated Morbidity Using the Plant-Based Repellent Zanzarin: A Randomized,

Controlled Field Study in Rural Madagascar. PLoS Negl Trop Dis. 2013; 7(9). doi: 10.1371/journal. pntd.0002426 PMID: 24069481

- 46. Thielecke M, Raharimanga V, Stauss-Grabo M, Rogier C, Richard V, Feldmeier H. Regression of severe tungiasis-associated morbidity after prevention of reinfection: a case series from rural Madagascar. Am J Trop Med Hyg. 2013; 89(5):932–6. doi: 10.4269/ajtmh.13-0244 PMID: 24043689
- 47. Norman C. Dominance statistics: Ordinal analyses to answer ordinal questions. Psychological Bulletin. 1993; 114(3):494–509. doi: 10.1037//0033-2909.114.3.494
- **48.** Romano J, Kromrey JD, Coraggio J, Skowronek J. Appropriate statistics for ordinal level data: Should we really be using t-test and cohen's d for evaluating group differences on the NSSE and other surveys? Annual meeting of the Florida Association of Institutional Research; Florida, 2006.
- **49.** Loft KE, Nissen MH. *Tunga penetrans* in a young dog imported to Denmark from Brazil; A case report. Vet Dermatol. 2009; 20(4):300–3. doi: 10.1111/j.1365-3164.2009.00765 PMID: 19659542
- Heukelbach J, Franck S, Feldmeier H. Therapy of tungiasis: a double-blinded randomised controlled trial with oral ivermection. Mem Inst Oswaldo Cruz, Rio de Janeiro 2004; 99(8):873–6. doi: 10.1590/ s0074-02762004000800015 PMID: 15761605
- 51. Cooper JE. An outbreak of *Tunga penetrans* in a pig herd. Vet Rec. 1967; 80:365–6. doi: <u>10.1136/vr</u>. 80.11.365 PMID: 4166051
- 52. Cooper JE. *Tunga penetrans* infestation in pigs. Vet Rec. 1976; 98:472. doi: <u>10.1136/vr.98.23.472</u> PMID: 936467
- Rietschel W. Observations on the sand flea (Tunga penetrans) in humans and dogs in French Guiana. Tierarztl Prax. 1989; 17(2):189–93. PMID: 2763291
- Ade-Serrano MA, Olomolehin OG, Adewunmi A. Treatment of human tungiasis with niridazole (Ambilhar): a double blind placebo-controlled trial. Ann Trop Med Parasitol. 1982; 76:89–92. doi: <u>10.1080/</u> 00034983.1982.11687508 PMID: 7044324
- 55. Cardoso A. Generalised tungiasis treated with thiabendazole. Arch Dermatol. 1981; 117:127. doi: 10. 1001/archderm.1981.01650030003001
- 56. Klimpel S, Mehlhorn H, Heukelbach J, Feldmeier H, Mencke N. Field trial of the efficacy of a combination of imidacloprid and permethrin against Tunga penetrans (sand flea, jigger flea) in dogs in Brazil. Parasitol Res. 2005; 97(SUPPL. 1):S113–S9. doi: 10.1007/s00436-005-1454-z PMID: 16228267
- 57. Hellmann K, Knoppe T, Krieger K, Stanneck D. European multicenter field trial on the efficacy and safety of a topical formulation of imidacloprid and permethrin (Advantix) in dogs naturally infested with ticks and/or fleas. Parasitol Res. 2003; 90 Suppl 3:S125–6. Epub 2003/08/21. doi: 10.1007/s00436-003-0912-8 PMID: 12928876.